Author’s response to reviews

Title: Advocacy of diagnostic criteria for maxillary incisive canal cysts based on alteration of normal maxillary incisive canals according to aging in Japanese populations

Authors:
Naruhiro Ueda (r16ueda@fa.kyu-dent.ac.jp)
Tatsurou Tanaka (t-tanaka@kyu-dent.ac.jp)
Masafumi Oda (r07oda@fa.kyu-dent.ac.jp)
Nao Wakasugi-Sato (r16wakasugi@fa.kyu-dent.ac.jp)
Shinobu Matsumoto-Takeda (s-takeda@kyu-dent.ac.jp)
Yuichi Miyamura (r16miyamura@fa.kyu-dent.ac.jp)
Takaaki Jyoujima (r16jyoujima@fa.kyu-dent.ac.jp)
Kouichi Kiyota (r19kiyota@fa.kyu-dent.ac.jp)
Kensuke Tsutsumi (r19tsutsumi@fa.kyu-dent.ac.jp)
Yasuhiro Morimoto (rad-mori@kyu-dent.ac.jp)

Version: 2 Date: 04 Oct 2019

Author’s response to reviews:

Dear Professor Stamm:

This letter is in response to your correspondence of October 2, 2019, concerning our manuscript (HAFM-D-19-00056). We were pleased to have received the your comments and are now submitting a revised version of “Advocacy of diagnostic criteria for maxillary incisive canal cysts based on alteration of normal maxillary incisive canals according to aging in Japanese populations”. We have corrected your suggestion about competing interests and authors' contributions.

We hope that you will now find the manuscript acceptable for publication in Head and Face Medicine, and we look forward to hearing from you at your earliest convenience.
Sincerely,

Yasuhiro Morimoto DDS PhD
Professor and Chairman
Division of Oral and Maxillofacial Radiology
Kyushu Dental University